Author’s response to reviews

Title: Gender differences in mental health problems among adolescents and the role of social support: results from the Belgian health interview surveys 2008 and 2013

Authors:
Filip Van Droogenbroeck (Filip.Van.Droogenbroeck@vub.be)
Bram Spruyt (Bram.Spruyt@vub.be)
Gil Keppens (Gil.Keppens@vub.be)

Version: 2 Date: 15 Dec 2017

Author’s response to reviews:

Dear Editor,

1. We renamed the Purpose section in the Abstract so that it appears as Background.

2. We added a 'Declarations' heading for our manuscript before the references section. We ensured that this section contains all of the sub-sections as described by the submission guidelines:

   https://bmcpsychiatry.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

3. Our revised copy has no tracked changes, colored or highlighted text.

Kind regards,